UNIVERSITY^{OF} BIRMINGHAM

University of Birmingham Research at Birmingham

Who decides 'what works', and how does it get implemented? Insights from the UK's new centre for implementing evidence in adult social care

Tawodzera, Obert; Glasby, Jon

DOI:

10.1332/23978821Y2024D000000034

License:

Creative Commons: Attribution (CC BY)

Document Version

Publisher's PDF, also known as Version of record

Citation for published version (Harvard):

Tawodzera, O & Glasby, J 2024, 'Who decides 'what works', and how does it get implemented? Insights from the UK's new centre for implementing evidence in adult social care', *International Journal of Care and Caring*. https://doi.org/10.1332/23978821Y2024D000000034

Link to publication on Research at Birmingham portal

General rights

Unless a licence is specified above, all rights (including copyright and moral rights) in this document are retained by the authors and/or the copyright holders. The express permission of the copyright holder must be obtained for any use of this material other than for purposes permitted by law.

- •Users may freely distribute the URL that is used to identify this publication.
- •Users may download and/or print one copy of the publication from the University of Birmingham research portal for the purpose of private study or non-commercial research.
- •User may use extracts from the document in line with the concept of 'fair dealing' under the Copyright, Designs and Patents Act 1988 (?)
- •Users may not further distribute the material nor use it for the purposes of commercial gain.

Where a licence is displayed above, please note the terms and conditions of the licence govern your use of this document.

When citing, please reference the published version.

Take down policy

While the University of Birmingham exercises care and attention in making items available there are rare occasions when an item has been uploaded in error or has been deemed to be commercially or otherwise sensitive.

If you believe that this is the case for this document, please contact UBIRA@lists.bham.ac.uk providing details and we will remove access to the work immediately and investigate.

Download date: 07. May. 2024

Print ISSN 2397-8821 • Online ISSN 2397-883X • https://doi.org/10.1332/23978821Y2024D000000034

Accepted for publication 20 March 2024 • First published online 17 April 2024



This article is distributed under the terms of the Creative Commons Attribution 4.0 International license (http://creativecommons.org/licenses/by/4.0/).

research article

Who decides 'what works', and how does it get implemented? Insights from the UK's new centre for implementing evidence in adult social care

Obert Tawodzera[®], o.tawodzera@bham.ac.uk Jon Glasby[®], j.glasby@bham.ac.uk University of Birmingham, UK

The UK's adult social care system faces severe challenges, including funding shortages, unmet needs and an overburdened workforce. Resultantly, there is a push for high-quality evidence in service enhancement and resource allocation. Using evidence to improve services is essential, but questions arise about best practices for identifying 'what works', integrating evidence into everyday practice and addressing resource constraints. Findings from a 2021 UK survey and consultative forums with stakeholders across adult social care underscore concerns about implementing evidence-based practices and highlight the need for increased collaboration to expand the evidence base. These findings shed light on stakeholders' perspectives regarding factors shaping adult social care practices, opportunities for evidence to play a greater role, definitions of valid evidence and priorities for change. Improved communication and coordination within the sector are crucial to enhance evidence-based decision making, focus limited resources on proven strategies and shape a more effective, evidence-informed adult social care system.

Keywords adult social care • evidence • implementation • lived experience • practice knowledge

To cite this article: Tawodzera, O. and Glasby, J. (2024) Who decides 'what works', and how does it get implemented? Insights from the UK's new centre for implementing evidence in adult social care, *International Journal of Care and Caring*, XX(XX): 1–20,

DOI: 10.1332/23978821Y2024D000000034

Introduction

Previous debates around the use of evidence in adult social care

Over several decades, there have been growing calls for the delivery of UK public services to be more 'evidence based'. This has acquired particular significance in the health service, where the evidence-based movement has been highly influential in

ensuring that new clinical interventions, equipment and medications are rigorously tested before use with patients and that practitioners are up to date with the latest clinical evidence. Politically, such thinking acquired particular significance under the New Labour governments of 1997–2010, with a stated desire to move away from traditional 'left versus right' debates about the state versus the market and to focus instead on 'what works' (Davies et al, 2000; Nutley et al, 2007; Boaz et al, 2019). While this led to a sense of optimism that evidence could play a greater role in shaping policy and practice, there were also significant limitations to this approach. As Boaz et al (2019: 6) caution:

There was, however, already a considerable amount of scepticism about the possibility of post-ideological politics in 2000. There were also concerns about the desirability of evidence-driven policies and practices, due to unease about the quality of the evidence base itself, and there were worries that the limitations of evidence as a basis for decision making were being underplayed. Evidence does not substitute for a continuing need to adjudicate between different sets of values (for example, on what matters), and there is an on-going need to exercise judgements that draw on experience and expertise alongside evidence.

These are important warnings, and many more recent actors and commentators refer to 'evidence-informed' policy and practice as a way of recognising that evidence should play a role in shaping public services but is far from the only factor (Monaghan and Ingold, 2019). This, for example, is the approach adopted by organisations such as Research in Practice, a membership organisation that brings together academic research, practice expertise and the experience of people accessing services to support evidence-informed practice in England. In Wales, Social Care Wales (2021) argues that 'evidence' should include formal research, the voices of people who use services and carers, and wisdom and knowledge from practitioners, organisations and policymakers, while practitioner-led initiatives like Developing Evidence Enriched Practice (DEEP) stress the importance of 'evidence-enriched practice'. In Scotland, the Institute for Research and Innovation in Social Services (Iriss, nd) believes that 'being able to access and use knowledge (research, practice wisdom and life story experience) well can help practitioners to learn and develop their practice; improve organisational culture; and contribute to changing the system'. In Northern Ireland, the Office of Social Services within the Department of Health has worked with the Social Care Institute for Excellence (SCIE) to make available a series of research, evidence and quality improvement resources (many of which combine insights from research, practice and lived experience), recognising that 'research and evidence underpin effective social work practice in Northern Ireland providing a knowledge base which supports social workers to inform and improve the services they provide' (SCIE, 2022). At the local level, various initiatives are seeking to develop a more evidence-informed culture, including recent research-practice partnerships funded by the National Institute for Health and Care Research (NIHR) and other similar 'researcher-in-residence' models (such as York's 'Curiosity Partnership' or the 'SCREEN' project in the Midlands Partnership NHS Foundation Trust [see Marshall et al, 2014; Gradinger et al, 2019; Wakefield et al, 2021]).

However, whether 'evidence based', 'evidence informed' or 'evidence enriched', most people would agree that public services, wherever possible, should be based on evidence of what is most effective when seeking to meet people's needs, albeit they would probably also argue that agreeing and understanding what is most effective is far from straightforward or unproblematic (Monaghan and Ingold, 2019; Oliver and Boaz, 2019). As a result, we have seen the development of national bodies like the National Institute for Health and Care Excellence (NICE), the SCIE and a series of 'what works centres' (covering topics like ageing, homelessness, early intervention, well-being, educational achievement, local economic growth, crime reduction and children's social care), all seeking to review evidence of what makes a difference and ensure that this is used to guide the work of front-line services and practitioners (see Gough et al, 2018). Although particularly associated (in the UK) with New Labour's 'Third Way', the desire to focus on 'what works' has continued since 2010 in a period of austerity, when many public services have faced significant financial, workforce and service pressures (see, for example, Glasby et al, 2020), and when there has had to be an overriding emphasis on using each pound of scarce public resource to most effect.

In this article, we adopt an inclusive definition of evidence that encompasses multiple forms of knowledge relevant to adult social care. Our constructivist approach emphasises the subjective and interpretive nature of reality (Lincoln and Guba, 2013). Therefore, we broadly define evidence to include insights derived from research, the lived experience of both people who draw on care and support and their carers, and the practice knowledge of adult social care staff. Within this, 'lived experience' refers to the first-hand, experiential knowledge gained by people with care and support needs and their unpaid caregivers. It encompasses both secondary evidence gathered through research with these groups and their direct participation in knowledge production activities, for example, through sharing their experiences and stories. 'Practice' or tacit knowledge refers to the skills, wisdom and know-how accumulated by social care practitioners through hands-on work and training in the field.

Incorporating lived experience, practice knowledge and academic research provides a more holistic understanding of evidence and of 'what works'. This aligns with our philosophical values regarding the nature of expertise and the validity of diverse voices and different ways of knowing the world. We aim to elevate the prominence of lived experience and practice knowledge as meaningful ways of understanding the world, not subordinate to academic research. However, we believe that all three are vital and can be brought together through a process of triangulation, consensus building and synthesising complementary insights. This creates a more comprehensive and inclusive foundation for designing and delivering high-quality adult social care services (Social Care Wales, 2021).

Barriers to the use of evidence

Although it is easy to say that services should be based on evidence of what works, translating evidence into practice is highly complex and requires an in-depth understanding of the broader social, political and economic context in which UK adult social care operates. While implementing evidence is challenging in

any public service, it is particularly difficult in adult social care, for multiple, interconnected reasons:

- The structure/nature of the adult social care sector: service planning and delivery, both within and across the four nations of the UK, is complex and fragmented and involves a diverse range of stakeholders (Glasby and Needham, 2020). Market-based mechanisms can mean that relationships are competitive and discourage the sharing of good practice, while widespread structural variations make it challenging to compare practice or outcomes across different areas (Cameron et al, 2014). There are few shared spaces for different people to come together and many disparate organisations, conferences and networks. The adult social care workforce, widely seen as underfunded, undervalued and overloaded (Dromey and Hochlaf, 2018; Manthorpe and Moriaty, 2021), lacks time, resources and support to engage with evidence or bring about change, and the costs of doing this can be prohibitive (especially for user-led organisations, carers, care workers and small providers) (Godar and Holmes, 2017; Lamont et al, 2020). When thinking about UK adult social care, moreover, there is also a risk that individual initiatives are too focused on one particular national context to be relevant or sensitive to other national contexts. Equally, there is a corresponding risk that we struggle to share learning across the four nations, failing to make the most of what might be seen as four 'natural experiments', each with key lessons to share with the others (Bell, 2010).
- Various cultural challenges: a tendency to develop small-scale pilots without mainstreaming or scaling up approaches that seem promising (SCIE, 2020); a reluctance to embrace learning from other local areas (the 'not invented here' syndrome); a societal failure to recognise the importance and complexity of care work (influenced by gendered assumptions about the nature of care and the role of women, as well as ageist/ableist assumptions about the value of the lives of people with care and support needs); a task orientation that can sometimes lead to a neglect of relationships and the human aspects of care (Yeandle and Buckner, 2017; MacLeavy, 2021); a lack of opportunities to develop a shared vision and agree on common values, making it hard for stakeholders to work towards common goals, confident of what success would look like (Ghate and Hood, 2019); a tendency to rely on 'hierarchies of evidence' that prioritise medical and natural science methods and undervalue other ways of knowing the world (Swinkels et al, 2002; Glasby and Beresford, 2006); and a reliance on passive or linear 'dissemination' models that ignore the realities of practice and fail to provide the practical support and receptive contexts that might help social care staff begin to work in different ways (Green, 2008).

Similar themes have also been identified by national bodies, such as Social Care Wales (2021), which has set out a series of guiding principles and more specific recommendations when seeking to overcome barriers to the greater use of research evidence (see Box 1).

Box 1: Social Care Wales's (2021) guiding principles to improve the use of evidence in social care

Guiding principles underpinning all recommendations:

- Relationships are central: interpersonal relationships and collaboration are important in enabling people to access, understand and use evidence.
- 2. Partner and collaborate: maximise collaboration between practice, policy, research and people with lived experience when designing services, undertaking research and developing and sharing evidence.
- Practicable and tangible: evidence, whether communicated in training, written summaries or through other methods, should provide practical and applicable knowledge and/or tools to help use by social care staff.
- 4. Take a whole-place approach: recognise barriers and facilitators across the system, including leadership, culture and practical/structural factors. Make links between local, regional and national levels.

Developing Improving Adult Care Together: the UK's centre for implementing evidence in adult social care

Against this background, this article reports findings from a national survey carried out across the UK adult social care sector in 2021, as well as themes from five national consultative bodies ('IMPACT assemblies') that met twice in 2021. These activities were part of a 'co-design' phase during the development of Improving Adult Care Together (IMPACT), the UK's new centre for implementing evidence in adult social care. The aim of IMPACT is to: increase the use of high-quality evidence, leading to better care practices, systems and outcomes; build capacity and skills in the adult social care workforce to work with evidence of different kinds in order to innovate and deliver better outcomes; develop relationships between a wide range of stakeholders across the sector to improve outcomes for people who draw on services and their families; and improve understanding of what elements of evidence implementation do and do not work in practice, using this to overcome barriers.

IMPACT believes that how we design and deliver adult social care can be improved by drawing on evidence of what works. However, as suggested earlier, 'evidence' should include insights from different types of research, from the lived experience of people using services and their carers, and from the practice knowledge of social care staff. Cognisant of the fragmented nature of adult social care, we believe that the sector stands the best chance of making a difference if stakeholders find ways to come together to work on common problems and solutions. We are well aware that different people have different levels of power and some voices are heard less often than others. Trying to hear as many different voices as possible and reduce traditional power imbalances is not only morally right; it also maximises the evidence/expertise available to the sector when trying to bring about positive change.

IMPACT's funders commissioned the new centre with three phases: a 'co-design' phase (April–December 2021); an 'establishment' phase (throughout 2022); and a

five-year 'delivery' phase (2023–27) – thereby building in time to consult the sector in all its diversity and across the UK as to exactly what IMPACT should do, what topics it should prioritise and how it should fit alongside other organisations and initiatives. As part of this process, we carried out a national survey and set up five 'IMPACT assemblies' across the UK's four nations (one each in Northern Ireland, Scotland and Wales, and two in England). Our assemblies and national survey included a mix of people, for example, those who draw on care and support, carers, front-line practitioners, service providers, commissioners, researchers, and representatives of national bodies, including people who are well networked with others across the social care system and those who have not had the opportunity to take part in such debates before (see the 'Methods' section). Key themes from the national survey, supplemented by additional insights from our assemblies, form the basis of this article.

Methods

National survey

The survey was conducted using an online form created on Qualtrics (web-based software that allows users to create/send out a survey, monitor response rates and collate results). We were interested in hearing from anyone involved with or connected to adult social care, particularly people who do not usually take part in such debates or whose voices are seldom heard. To maximise the scope for different people to take part in ways that worked for them, we made sure that people could engage with us in different ways:

- The survey was available in an easy-read version/large-print version (designed for respondents with a learning disability and/or people with visual impairment).
- The survey was translated into Welsh and the five most common other languages (Polish, Bengali, Urdu, Punjabi and Gujarati).
- People could complete the survey by themselves or with support from someone else (for example, a group of self-advocates with learning disabilities could talk through the survey questions as a group, with a support worker submitting their views).
- People could complete the survey as an individual or as part of a group (for example, a chair could talk through the questions at an annual general meeting, or a support worker could complete the survey with a group of self-advocates with learning disabilities).
- People could contact us to ask for a hard copy or send in responses by email
 if this would work better for them.

The survey was distributed online via social media, in regular bulletins via the (very extensive) networks of IMPACT's leadership team and consortium (which include a broad range of national policy and practice partners, with very extensive reach into all parts of the sector), and via adult social care stakeholders across the UK. A lived experience engagement lead also helped to promote the survey to user- and carerled organisations and various national co-production networks. The survey was also promoted in individual stakeholder engagement meetings with over 100 social care organisations across the UK and via sessions at a series of local, regional and national social care events and conferences.

The survey asked about: people's position within the adult social care sector, their geographical location and a number of protected characteristics; possible ways in which IMPACT should prioritise its work; possible topics that IMPACT could usefully explore; what role (if any) evidence currently plays in shaping adult social care; ways of boosting the contribution of evidence/overcoming current barriers; what constitutes valid evidence/whose voice should be heard during these debates; and how people like to communicate. While we have used insights from our findings to design IMPACT's delivery models and approaches to communication, the rest of this article focuses on the insights that we received into what shapes what happens in adult social care, the role of evidence, barriers to greater use of evidence and what might help.

The survey was primarily quantitative, asking people to state the importance of a series of possible answers/topics by ranking them from 0 (not important at all) to 5 (very important). However, several questions also enabled people to add their own priorities and ideas, for example, asking: 'Are there other things that matter to you? (Please specify and explain your answer)'. The survey was preceded by clear information explaining the purpose of the survey, that participation was voluntary, that participants would not be identifiable and what the results would be used for. We did not want to put people off by insisting that they complete all questions, so people could choose which questions they wanted to answer. This means that not all questions have been completed by all respondents, so the number of responses that were analysed for each question was lower than the total number of survey responses (with the sample size for each question set out in our findings section).

Data analysis

The data collected from respondents were analysed using descriptive statistics and frequency counts, as well as a process of thematic analysis for the qualitative data provided by respondents to open-ended or free-text questions. While we have data for each question that can be broken down by the different nations of the UK and by people's roles and positions within the social care system, key themes were remarkably consistent across all groups and locations (for further discussion, see later). As a result, this article does not break each question down in this way (as this would lead to lots of extra charts with large amounts of detail, though essentially repeating the same overall messages), choosing instead to highlight any minor differences of emphasis in the main text.

IMPACT assemblies

A key feature of IMPACT's work is its 'assemblies' (two in England and one each in Wales, Scotland and Northern Ireland), each of which consisted of around 30–35 people, including those who draw on care and support, carers, practitioners, providers, commissioners, researchers, and broader stakeholders, who took part in a series of deliberative forums to: identify and build consensus around IMPACT's priorities; ensure that IMPACT's work is embedded in and responsive to the different policy and practice contexts of the four nations; debate and refine proposed delivery models; provide feedback on any early impact; support the scaling up of effective evidence-based practice and promote wider cultural change; and understand emerging issues

and new developments in adult social care. These are similar to the citizens' assembly convened for the Health and Social Care Select Committee and the Housing, Communities and Local Government Select Committee during their inquiry into adult social care funding (INVOLVE, 2018), which involved a three-step process (learning, deliberation and decision making), with facilitators to ensure everyone was heard and felt comfortable participating.

While the 'learning' phase was different for our assembly members (who are already experts in different aspects of adult social care), it still involved learning from each other, so similar design features were relevant. For example, techniques included: sending out material and questions in advance so that people could prepare with whatever support works best for them; all discussion taking place in facilitated small groups to help as many people as possible feel comfortable speaking up; lots of opportunities to ask questions and clarify if someone did not understand some of the materials; and regularly mixing up groups to expose people to a range of views/prevent dominant narratives. People who felt more comfortable listening to the discussion and contributing their individual thoughts by email or phone afterwards were welcome to do so. Our approach also has similarities to insights from the James Lind Alliance (2018) method for agreeing on priorities between National Health Service (NHS) clinicians and patients, the Harvard 'Tobin project process' used in the Economic and Social Research Council (ESRC) UK Collaborative Centre for Housing Evidence (Gibb, 2018), and the Scottish Learning Disabilities Observatory's 'citizens' juries' (SLDO, 2022).

Each assembly met twice during 2021, with two half-day workshops. They were co-chaired by a lead university and a policy/practice partner (either from the leadership team or from IMPACT's broader consortium and partners), ensuring that different groups and voices were represented across the five assemblies. Meetings were held online using Zoom video-conferencing services. Every step was taken to minimise the risk of digital exclusion, including providing training/guidance in advance and working with individuals with particular access needs to make sure that they felt comfortable participating in this way. Feedback indicated the viability of Zoom as a means of enabling meaningful participation, including for people who might otherwise have been unable to attend a central, face-to-face meeting (for further discussion, see Archibald et al, 2019). Where people were not participating as part of their 'day job', they were paid for their time using standard NIHR rates (this was mainly for people who draw on care and support, carers, front-line care workers not on permanent contracts or not supported to attend by their employer, and some small care providers).

Data analysis

Each assembly included two small group exercises, with a facilitator feeding back the top three points from each group in plenary and submitting more detailed notes of the discussion to the co-chairs. The latter then produced a report on each of their assembly discussions, with key themes compiled across both meetings of all five assemblies (that is, ten in total). Typically, our survey data provided a broad overview of people's experiences, perspectives and priorities, while more in-depth discussion and debate in our assemblies enabled us to sense-check emerging themes, explore the possible rationales behind them and generate additional nuance and richness of insight.

Insights from our assemblies were captured using a thematic analysis approach, as outlined by Braun and Clarke (2006), involving familiarisation with the data, generating

initial codes, searching for patterns to develop overarching themes that encapsulate recurring ideas and constantly checking back to make sure that the emerging themes continue to represent the data. Robustness was enhanced through triangulation with the survey data and by overarching themes being worked up together by two members of the research team (X and Y), initially on an independent basis, then meeting to combine emerging insights, agree on a single approach and check the consistency of understanding and interpretation. In particular, assemblies tended to break up into small and very diverse groups to enable people to feel more comfortable participating than in plenary, and members of the IMPACT team helped agree on key themes to feed back, took notes and recorded key areas of consensus. This means that we do not have the detailed backgrounds of the individuals who are quoted in the findings section later, albeit such quotes are chosen because they illustrate clear and strongly held views rather than a perspective shared by one group of people but not by another. As with our survey, there was significant consensus between different groups, so this approach to collating and reporting overall themes/building consensus – alongside insights from our national survey – was felt to be appropriate and helpful in identifying shared priorities. It was also felt to boost participation, as a number of people may not have been comfortable sharing their views if individual quotes were being recorded as from specific groups of people (rather than just as illustrations of a general theme).

Ethics

UK social care research is governed by strict regulations to ensure high ethical standards set by an organisation known as the Health Research Authority, supplemented by ethical approval panels in individual universities (full details of the Health Research Authority are available at the NHS Health Research Authority website). Since IMPACT is an implementation centre, not a research centre, its stakeholder engagement activities were deemed to be a form of service development or audit, not requiring Health Research Authority or university research ethics approval – though still needing to be conducted in an ethical and appropriate manner. This was reviewed and confirmed by the University of Birmingham's research governance team. To ensure that the insights in this article were collected in an ethical way, people completing the survey had an initial briefing note on the way in which data would be used, as follows:

We don't need your name, so all responses will be anonymous. You can provide contact details at the end if you'd like to keep in touch, but these won't be linked to your answers. We will ask a few questions about you at the end of the survey. This is to make sure that we are talking to a wide range of people. You don't have to answer any of these questions if you don't want to. We will use data from this survey to shape our work, and we may write articles, blogs and reports about what we find – but no one will be able to identify you from anything we write.

People applying to be members of our assemblies received a background briefing and signed to agree with the statement: 'I understand that what I say will be used in a non-attributable way, but that I may be named as a member of an IMPACT assembly, and that discussions from assemblies will be written up in various different reports, articles and blogs.'

Results

Participant characteristics

A total of 2,165 people completed our survey, making it (we think) one of the largest surveys ever to be undertaken on this kind of topic in UK adult social care. As but one example, when the government's independent review of children's social care (in England) issued a call for advice (which, similar to our survey, asked how the review should work and the questions it should focus on) and a linked call for evidence, the two calls received just over 1,100 responses (McAllister, 2022). In Scotland, the Feeley Review (Scottish Government, 2021) of adult social care commissioned by ministers in response to COVID-19 engaged just over 1,000 people (including 228 written submissions, 13 engagement events and 128 meetings with stakeholders). In Northern Ireland, the consultation on the reform of adult social care engaged over 375 participants in 17 online engagement events and received over 200 responses through an online survey and written responses (Department of Health, 2023).

A total of 84 per cent of respondents completed the survey as individuals, 10 per cent completed on behalf of a group and 6 per cent completed on behalf of someone else. From the free text added to these responses, most of these groups seemed to be people in supported living, where a group responded with support from staff. The largest number of respondents were from England (79 per cent of the total number of respondents), with Scotland at 9 per cent, Wales at 8 per cent and Northern Ireland at 3 per cent (the other 1 per cent were people whose work covers all the four nations or those who are based in other locations, such as the Channel Islands). This distribution strongly mirrors the general population of the UK, though with a slightly higher response rate per head of population for Wales and slightly lower for England (ONS, 2021). The vast majority of surveys returned were in English. Only seven were returned in Welsh, four in Bengali and three each in Polish, Urdu, Gujarati and Punjabi. A total of 63 of the returned questionnaires were of the easy-read version.

As illustrated in Table 1,80 per cent of respondents were female, which is consistent with the make-up of the overall social care workforce (for data on England, see, for example, Skills for Care, 2021), albeit this raises important questions about gender and future scope to get men more involved in caring roles. In terms of ethnicity, 92 per cent of respondents were White, 3 per cent were Asian/Asian British, 2 per cent were Black or Black British, 2 per cent were from a mixed ethnic group (for example, White and Black Caribbean, White and Black African, White and Asian, and so on) and 1 per cent were from another background (for example, Arab). This is slightly different to the overall make-up of the UK population, where around 87 per cent of people are white, indicating a need for IMPACT to do more to engage people from black and minority ethnic communities (in response, one of our first projects in 2022 has focused on personalisation and people from black and minority ethnic communities). In terms of other protected characteristics, 90 per cent of respondents said that they were heterosexual, and 27 per cent said that they had a disability or long-term health condition.

When describing their role within adult social care, respondents came from a broad range of backgrounds (see Table 2), including front-line practitioners (25 per cent), carers (13 per cent) and people who draw on care and support (9 per cent). This means that just under half of respondents were people who draw on

Table 1: Our respondents

	Frequency	%
Nation (n = 1,609)		
Scotland England	1,269	79
Scotland	137	8.5
Northern Ireland	51	3.1
Wales	132	8.2
Other	20	1.2
Sex (n = 1,281)		
Female	1,024	80
Male	257	20
Sexual orientation (n = 1,208)		
Heterosexual	1,089	90.1
Gay	17	1.4
Lesbian	36	3.0
Bisexual	35	2.9
Other	31	2.6
Ethnicity (n = 1,275)		
White	1,171	91.8
Mixed	24	1.8
Asian/Asian British	36	2.9
Black/Black British	27	2.0
Other	18	1.4
With a form of disability (n = 1,273)		
Yes	344	27
No	929	73

Table 2: People's role within adult social care

	Fre- quency	%
Currently draw on care and support	137	8.5
Carer for an adult who needs support (on an unpaid basis)	203	13.1
'Front-line' worker in adult social care	391	24.6
Manager or owner of a care service (a 'service provider')	265	16.7
Organise, fund and/or commission services (a commissioner or strategic planner)	77	4.8
Director of social services (or equivalent)	13	0.8
Work in the NHS	42	2.6
Work in housing	18	1.2
Local councillor	3	0.2
Work for a national voluntary organisation that advocates and campaigns	48	3.2
Work for a regulator	9	0.6
National policymaker or work in a national body	29	1.9
Work in another role please specify	349	21.9

Note: n = 1,584.

care and support, carers, and front-line workers, groups that are often seldom heard directly in key policy debates. Where people said that they worked in another role (21.9 per cent, n = 349), they cited such backgrounds as research, occupational therapy, consultancy, advocacy and working for non-governmental organisations. In practice, we recognise that a number of these descriptions may overlap and that people often have multiple roles/identities. In answering this question, people could select multiple roles or perspectives, recognising that people may have a number of different identities; therefore, the overall total may differ from some other tables.

What shapes adult social care in the UK?

To help us understand more about the role of evidence, we asked respondents what they think shapes what happens in adult social care (see Table 3). The majority (83 per cent; n = 1,035) believed that lack of funding was the key factor. Other key influences were national policy (selected as very important by 42 per cent, n = 521), local policy (selected as very important by 37 per cent, n = 464), what service providers want to

Table 3: What shapes what happens in social care?

	0	1	2	3	4	5
Funding (how much money is available to pay for social care)	0.5%	1.2%	1.8%	3.8%	10.1%	82.7%
National policy (what national governments say should happen)	1.9%	3.7%	8.3%	19.1%	25.3%	41.7%
Local policy (what local councils say should happen)	1.3%	3.1%	7.9%	18.4%	31.9%	37.4%
What service providers want to deliver	3.7%	9.1%	15.1%	25.1%	21.6%	25.4%
What the local manager and senior leader think is best	4.2%	8.2%	16.4%	25.2%	25.7%	20.3%
What people who use services want	14.1%	23.6%	21%	17.4%	7.6%	16.3%
What (unpaid) carers want	21.6%	27.9%	18.5%	14.1%	6.3%	11.6%
Evidence of what works	8.8%	20%	27.9%	22.4%	7.3%	13.5%
Things are done the same way they have always been done	8.7%	10.3%	16.5%	19.6%	21.2%	23.6%
Good practice from another local area	7.5%	15.9%	25%	25.9%	15.9%	9.8%
Good practice from another sector (for example, the NHS)	9.6%	20.7%	26%	22.2%	11.4%	10.1%
International good practice	21.3%	25.4%	24.4%	12.7%	7.2%	9.1%

Notes: Each scored on a 0–5 scale, with '5' for 'shapes social care a lot' and '0' for 'does not shape social care at all'. n = 1,251.

deliver (seen as very important by 25 per cent, n = 312) and things being done in the same way they always have (23 per cent, n = 287). Relatively few people felt that social care is shaped by good practice in other sectors or international good practice. What people drawing on care and support or carers want was also given a low score. Crucially for IMPACT, evidence of what works was only seen as very important in shaping social care by 13 per cent (n = 166) of respondents and as important by 7 per cent (n = 90), potentially a sad indictment of the progress of the evidence-based practice movement.

What constitutes valid evidence?

When considering what kinds of knowledge are useful in identifying what works, many respondents believed that people's lived experiences of drawing on care and support or of being a carer should be considered a crucial source of evidence (see Table 4). Respondents also felt that the practice knowledge of social care staff was an important source of expertise. While research was seen as important too, it received a lower score than these other forms of knowledge. These findings received strong endorsement by our assembly members, who also believed that evidence of what works should include different kinds of research, the lived experiences of both people who draw on care and support and their carers, and the practice knowledge of front-line workers. Although Assembly members felt that all three of these ways of knowing the world were important, they felt that particular types of research (especially quantitative/ economic forms of research) were often prioritised by decision makers ahead of other sources of evidence. While they wanted IMPACT to bring together and triangulate each of these forms of knowledge, they suggested that we would need to 'turn the volume up' on lived experience and practice knowledge if they were going to be seen as just as important a form of expertise as research.

Table 4: Understanding what works might involve different kinds of knowledge. How important are each of these to you?

	0	1	2	3	4	5
Research evidence based on studies and evaluation	1.4%	2.4%	5.4%	20.8%	30.1%	40%
People's lived experiences of using services or providing care	0.2%	0.3%	1.3%	3.5%	13.5%	81.1%
The practical know-how of adult social care staff	1%	1.8%	3.4%	10.7%	25.2%	57.9%

Notes: Each scored on a 0-5 scale, with '5' for 'very important' and '0' for 'not important at all'. n = 1,251.

What would help ensure that evidence about what works in adult social care is used?

As set out in Table 5, the majority of respondents would value: funding to enable the participation of people who draw on care and support, carers, and front-line staff; practical support to make changes on the ground in the realities of local practice;

Table 5: What would help ensure evidence about what works in adult social care is used?

	0	1	2	3	4	5
More evidence	2.3%	2.5%	7.4%	23.3%	25.5%	38.9%
Practical support to change social care in your area	0.8%	0.8%	3.2%	10.8%	26.4%	58.0%
Funding to involve people who use services, carers and care workers	0.4%	0.8%	2.6%	7.7%	23.7%	64.8%
Easy-read, accessible guidance	1.3%	2.3%	8.1%	18.4%	27.0%	43.0%
Opportunities for people to work on practical changes together	0.2%	0.9%	3.3%	10.8%	27.7%	57.1%
Free training materials (on what would work better)	1.4%	2.7%	9.1%	18.5%	26.1%	42.3%
Free training materials (on how to use evidence)	1.7%	3.0%	9.0%	20.0%	26.7%	39.7%

Notes: Each scored on a 0-5 scale, with '5' for 'very important' and '0' for 'not important at all'. n = 1,318.

and opportunities for people to come together from different backgrounds and parts of the system to work on practical changes together. Interestingly, 'more evidence' or free training (either on making better use of evidence or on what would work better) were less of a priority.

More generally, analysis of respondents' in-depth comments and discussions from assemblies shows that there is a consensus that there is enough evidence of what works but a lack of evidence that is co-produced by different stakeholders and a lack of shared vision about what success would look like across a fragmented sector:

We also need to think about how we create a dialogue between all stakeholders – currently, the dialogue in adult social care is hierarchical and bureaucratic, but we need to get everyone around the table and start to ask: what are we trying to achieve, what evidence is there, how do we work with it in practice, what difference can it make? (Assembly member)

Some also felt that current evidence is very 'academic' and not sufficiently accessible to people drawing on care and support, carers, or front-line workers. Many people were clear that we already know quite a lot about what works but the difficulty is doing it in practice: 'There're loads of evidence about what works – less about how what works in one situation or place can work in another place or situation. So, focus on making change rather than producing more and more evidence that people don't use' (survey respondent – public sector registered manager [service provider]).

Barriers to using evidence

When reflecting on the barriers to using evidence in adult social care, respondents highlighted such issues as a lack of funding, a lack of time and thinking space, a lack of priority given to adult social care in wider society, and a lack of shared vision (see Table 6).

Table 6: What are the barriers to using evidence to improve social care?

	0	1	2	3	4	5
Lack of funding	1.1%	1.9%	3.0%	11.0%	18.0%	65%
Lack of time and thinking space	0.9%	1.1%	2.9%	12.1%	25.1%	58.0%
No evidence exists	13.4%	12.8%	21.2%	26.5%	15.5%	10.6%
Evidence is not in an easy format	6.2%	8.1%	18.3%	27.9%	23.6%	15.8%
No priority given to social care in wider society	1%	1.6%	5.2%	11.6%	23%	57.6%
Local politics	1.1%	4.3%	12%	20.6%	25.3%	36.7%
National politics	1.2%	2.9%	8.2%	14.1%	23.1%	50.5%
Limited skills in applying evidence	2.1%	5.7%	15.4%	28.6%	25.2%	23.1%
No shared vision about what social care is about	2%	2.9%	7%	13.4%	22.8%	52%

Notes: Each scored on a 0–5 scale, with '5' for 'a very important barrier' and '0' for 'not a barrier at all'. n = 1,255.

Discussions in our assemblies also pointed to the fragmented nature of adult social care, feeling that a lack of shared spaces makes it hard for people to come together to reflect, learn and work together on what changes they want to see and what might work best. They also felt that the current system can be very hierarchical, with power tending to reside in people who might be a long way away from the issues at stake, leading to the risk that they then misunderstand these issues or do not try to solve the right problems:

There is a structural problem in the social care system. In most cases, people in offices do not know what is happening on the ground. I would propose a bottom-up approach with carers and care workers: those at the bottom leading the work and telling those at the top what is more important and what needs to be done, and then those at the top assisting. At present, it's those at the top who tell those at the bottom what to do, yet they do not have the experience that those at the bottom have. (Assembly member)

This was also felt to be an issue with a lot of academic research and with the work of some existing evidence centres, with a perception that some liked to research things that interested them or to make statements about the nature of the evidence, without necessarily fully understanding the realities of people's lives and front-line practice.

Another barrier identified by assembly members was that most evidence is not easily accessible by particular groups of people, including people who draw on care and support, their carers, and front-line workers, either because the evidence is not written with these groups in mind or because it is behind a paywall. For assembly members, anything that helps evidence to be more accessible would be a good thing.

Discussion and conclusions

Despite ongoing and increasing attempts to ensure that public services are based on the best available evidence, lots of barriers remain. Respondents to our survey and assembly members have been clear with us that evidence of what works is much less influential in determining what happens in adult social care than funding, national policy, local politics, what service providers want to deliver and how things have been delivered in the past. Depressingly, what people who draw on care and support or carers want, evidence of what works, and good practice from elsewhere seem to have much less impact on what actually happens in practice.

However, respondents have been equally clear about what they believe would help: funds to enable the participation of people who draw on care and support, carers, and front-line staff; practical support to make positive changes in the realities of local services; opportunities to work together on common problems and common solutions; a commitment to co-production and collaboration; and an explicit commitment to combining and valuing insights from research, lived experience and practice knowledge. The latter resonates with a constructivist approach to knowledge, where the subjective and interpretive nature of reality is paramount (Lincoln and Guba, 2013). These are all principles and ways of working that we have embedded throughout the work of IMPACT.

Although our study was carried out to inform our work as an evidence implementation centre, our findings may have implications for the sector as a whole and for debates about the implementation of evidence in other sectors. Although many people would argue that public services should be more 'evidence based' than has historically been the case, Glasby and Beresford (2006: 269) have claimed that 'this is a statement of a dilemma not a practical blueprint for the way forward'. While services should be based on what we know/think will work best, this raises key questions about what constitutes valid evidence, who decides, whose voice is heard when we explore different options for reform and how we actually get the subsequent evidence used in practice in the realities of local delivery and people's lives. Prior work in disability studies, user-controlled research and feminist theory underscores that evidence considered 'valid' often stems from dominant groups while excluding the experiential expertise of marginalised communities (Collins, 2002; Glasby and Beresford, 2006; Beresford, 2019). We must interrogate whose interests determine research agendas, whose voices are heard in exploring reforms and how to centre the knowledge of those facing structural oppressions. Unless inquiry aims, processes and applications are co-designed with both those who draw on care and support and carers, they risk perpetuating exclusion and maintaining the status quo.

Research grounded in diverse lived realities, not just external ideals, is key to transforming systems of inequity. This also links to Fricker's (2007) notion of 'epistemic injustice, in which particular people and groups can be excluded and silenced through a twin process of testimonial and hermeneutic injustice (that is, distrusting someone's word or ignoring them because of who the person is, as well as misunderstanding people's experiences because of being too distant from the reality of those experiences and/or where historical exclusion prevents people themselves from being able to understand and voice their own experiences).

As Nutley et al (2013) caution, moreover, there is no straightforward 'answer' as to what constitutes good evidence; rather, it depends on what we want to know and how we plan to use that evidence. They caution against expecting standards of evidence to heavily influence complex decision making by policymakers, service managers and local practitioners. Acknowledging the complexities and political nature of these processes, it is essential to appreciate the limited impact of evidence standards on practical outcomes in local delivery and people's lives.

Ghate and Hood (2019) indicate that social care encompasses intricate, multilayered challenges, rendering simple solutions to such complex issues wholly inadequate. Implementing evidence in practice is not therefore a technical matter of carrying out 'good research' and making it more easily available or training people to be able to use it. Rather, as Nutley et al (2007) contend in their examination of evidence-based policy and practice in public services, it also requires contending with myriad barriers embedded in organisational cultures and systems. Genuine implementation in adult social care is a much messier and inevitably more contested matter of hearing and valuing different voices, triangulating insights from different ways of knowing the world, and working alongside people in the realities of local services to provide practical support. In one of our assemblies, a participant described this as 'not just talking about it but rolling up your sleeves and getting your hands dirty', and this feels like an important metaphor for the work that IMPACT will undertake across the UK and for the skills and approaches that are needed to overcome the long-standing barriers to the greater use of evidence in adult social care summarised earlier.

Limitations

This scoping study has several limitations stemming from its preliminary nature and its purpose in informing the design of a new implementation centre. While the mixed-methods approach provides helpful insights, the survey sample was not stratified and the diverse but small-scale assemblies limit generalisability across diverse contexts in a scientific/statistical sense. However, as discussed earlier, the consensus that emerged from and between both forms of data collection perhaps suggests a common experience and generalisability of a different sort.

We were also very aware that our data were collected during the COVID-19 pandemic, which may have influenced participation and responses in unknown ways, though virtual data collection may well have helped mitigate this. While lived experience and practice knowledge provide vital complementary sources of insight alongside academic research, they too have limitations. Thus, some would argue that lived experience risks excluding the diversity of perspectives and relies on subjective recall, while practice knowledge may perpetuate outdated or harmful practices if not balanced with external accountability and formal research evidence. However, taken together, evidence from research, lived experience and practice – particularly when care is taken to achieve a significant depth and breadth of contributions, to reduce barriers to taking part, and to include seldom-heard voices – can nonetheless provide important insights to guide future policy, practice and implementation.

ORCID iD

Obert Tawodzera https://orcid.org/0000-0002-0386-0335 Jon Glasby https://orcid.org/0000-0003-3960-7988

Note

https://www.hra.nhs.uk/planning-and-improving-research/policies-standards-legislation/ uk-policy-framework-health-social-care-research/

Funding

This article draws on insights from a project funded by the ESRC through the UK Research and Innovation Strategic Priorities Fund (SPF) and the Health Foundation (ESRC Grant Number ES/V001035/1). The views here are those of the authors and are not necessarily representative of our funders or other IMPACT colleagues.

Acknowledgements

We would like to thank Professor Catherine Needham for very helpful comments on an earlier draft of this article. We would also like to thank the reviewers and editors of the journal for their comments and suggestions.

Conflict of interest

The authors declare that there is no conflict of interest.

References

- Archibald, M.M., Ambagtsheer, R.C., Casey, M.G. and Lawless, M. (2019) Using Zoom videoconferencing for qualitative data collection: perceptions and experiences of researchers and participants, *International Journal of Qualitative Methods*, 18: 1–8. doi: 10.1177/1609406919874596
- Bell, D. (2010) The Impact of Devolution: Long-Term Care Provision in the UK, York: Joseph Rowntree Foundation.
- Beresford, P. (2019) Public participation in health and social care: exploring the co-production of knowledge, *Frontiers in Sociology*, 3(41). doi: 10.3389/fsoc.2018.00041
- Boaz, A., Davies, H., Fraser, A. and Nutley, S. (eds) (2019) What Works Now? Evidence-Informed Policy and Practice, Bristol: Policy Press.
- Braun, V. and Clarke, V. (2006) Using thematic analysis in psychology, *Qualitative Research in Psychology*, 3(2): 77–101. doi: 10.1191/1478088706qp063oa
- Cameron, A., Lart, R., Bostock, L. and Coomber, C. (2014) Factors that promote and hinder joint and integrated working between health and social care services: a review of research literature, *Health and Social Care in the Community*, 22(3): 225–33. doi: 10.1111/hsc.12057
- Collins, P.H. (2002) Black Feminist Thought: Knowledge, Consciousness, and the Politics of Empowerment, New York: Routledge.
- Davies, H., Nutley, S. and Smith, P. (2000) What Works? Evidence-Based Policy and Practice in Public Services, Bristol: Policy Press.
- Department of Health (2023) Consultation on the Reform of Adult Social Care: Summary and Evaluation of Responses to Public Consultation, Belfast: Department of Health.
- Dromey, J. and Hochlaf, D. (2018) Fair Care: A Workforce Strategy for Social Care, London: IPPR.
- Fricker, M. (2007) *Epistemic Injustice: Power and the Ethics of Knowing*, Oxford: Oxford University Press.
- Ghate, D. and Hood, R. (2019) Using evidence in social care, in A. Boaz, H. Davies, A. Fraser and S. Nutley (eds) *What Works Now? Evidence-Informed Policy and Practice*, Bristol: Policy Press, pp 89–109.
- Gibb, K. (2018) Taking Co-production Seriously: Establishing the ESRC UK Collaborative Centre for Housing Evidence (CACHE), Glasgow: CACHE.

- Glasby, J. and Beresford, P. (2006) Who knows best? Evidence-based practice and the service user contribution, *Critical Social Policy*, 26(1): 268–84. doi: 10.1177/0261018306059775
- Glasby, J. and Needham, C. (2020) The neglect of adult social care during COVID-19, *British Medical Journal*, 370: m3103. doi: 10.1136/bmj.m3103
- Glasby, J., Zhang, Y., Bennett, M. and Hall, P. (2020) A lost decade? A renewed case for adult social care reform in England, *Journal of Social Policy*, 50(2): 406–37.
- Godar, R. and Holmes, D. (2017) The Use of Research Evidence Regarding 'What Works' in Local Authority Child Protection Systems and Practice, London: Early Intervention Foundation, Research in Practice and Local Government Association.
- Gough, D., Maidment, C. and Sharples, J. (2018) UK What Works Centres: aims, methods and contexts, https://eppi.ioe.ac.uk/cms/Default.aspx?tabid=3731.
- Gradinger, F., Elston, J., Asthana, S., Martin, S. and Byng, R. (2019) Reflections on the researcher-in-residence model co-producing knowledge for action in an integrated care organisation: a mixed methods case study using an impact survey and field notes, *Evidence & Policy*, 15(2): 197–215. doi: 10.1332/174426419x15538508969850
- Green, L.W. (2008) Making research relevant: if it is an evidence-based practice, where's the practice-based evidence?, *Family Practice*, 25(1): i20–4. doi: 10.1093/fampra/cmn055
- INVOLVE (2018) Citizens' Assembly on Social Care: Recommendations for Funding Adult Social Care, London: INVOLVE and House of Commons.
- Iriss (Institute for Research and Innovation in Social Services) (nd) About us, www. iriss.org.uk/about.
- James Lind Alliance (2018) *The James Lind Alliance guidebook: Version 8*, Southampton: JLA, www.jla.nihr.ac.uk.
- Lamont, T., Allen, R. and Geoghegan, L. (2020) Enabling Better Use of Evidence in Social Work Policy and Practice: A BASW/NIHR Roundtable Discussion, Birmingham: BASW and NIHR.
- Lincoln, Y.S. and Guba, E.G. (2013) *The Constructivist Credo*, New York: Routledge. MacAlister, J. (2022) The independent review of children's social care: final report, www.The-independent-review-of-childrens-social-care-Final-report.pdf.
- MacLeavy, J. (2021) Care work, gender inequality and technological advancement in the age of COVID-19, *Gender, Work & Organisation*, 28: 138–54.
- Manthorpe, J. and Moriarty, J. (2021) Workforce ethnic diversity in older people's care services: thinking back and thinking ahead in COVID-19 times, *Working with Older People*, 25(2): 170–8. doi: 10.1108/wwop-12-2020-0061
- Marshall, M., Pagel, C., French, C., Utley, M., Allwood, D., Fulop, N., Pope, C., Banks, V. and Goldman, A. (2014) Moving improvement research closer to practice: the researcher-in-residence model, *BMJ Quality & Safety*, 23: 801–5.
- Monaghan, M. and Ingold, J. (2019) Policy practitioners' accounts of evidence-based policy making: the case of Universal Credit, *Journal of Social Policy*, 48(2): 351–68. doi: 10.1017/s004727941800051x
- Nutley, S., Walter, I. and Davies, H. (2007) *Using Evidence: How Research Can Inform Public Services*, Bristol: Policy Press.
- Nutley, S., Powell, A. and Davies, H. (2013) What Counts as Good Evidence? Provocation Paper for the Alliance for Useful Evidence, St Andrews: Alliance for Useful Evidence/University of St Andrews.

- ONS (Office for National Statistics) (2021) Population estimates for the UK, England, Wales, Scotland and Northern Ireland: mid-2021, www.ons.gov.uk/peoplepopulationandcommunity/populationandmigration/populationestimates/bulletins/annualmidyearpopulationestimates/mid2021.
- Oliver, K. and Boaz, A. (2019) Transforming evidence for policy and practice: creating space for new conversations, *Palgrave Communications*, 5: 60. doi: 10.1057/s41599-019-0266-1
- SCIE (Social Care Institute for Excellence) (2020) Scaling Innovation in Social Care: Rapid Pragmatic Evidence Review, London: SCIE.
- SCIE (2022) Research, evidence and quality improvement, www.scie.org.uk/northern-ireland/research-evidence-quality-improvement.
- Scottish Government (2021) Adult social care: independent review, www.gov.scot. Skills for Care (2021) *The State of the Adult Social Care Sector and Workforce in England*, 2021, Leeds: Skills for Care.
- SLDO (Scottish Learning Disabilities Observatory) (2022) What is a citizens' jury?, www.sldo.ac.uk/inclusive-research/research-voices-project/what-is-a-citizens-jury/.
- Social Care Wales (2021) Using Evidence in Social Services and Social Care in Wales, Cardiff: Social Care Wales.
- Swinkels, A., Albarran, J.W., Means, R.I., Mitchell, T. and Stewart, M.C. (2002) Evidence-based practice in health and social care: where are we now?, *Journal of Interprofessional Care*, 16(4): 335–47. doi: 10.1080/1356182021000008265
- Wakefield, J., Lavender, S., Nixon, K., Hornby, S., Dobel-Ober, D., Lambley-Burke, R., Spolander, G., Bonner, P., Mallen, C. and Campbel, P. (2021) Social work and social care: mapping workforce engagement, relevance, experience and interest in research, *British Journal of Social Work*, 52(4): 2291–311. doi: 10.1093/bjsw/bcab169
- Yeandle, S. and Buckner, L. (2017) Older workers and care-giving in England: the policy context for older workers' employment patterns, *Journal of Cross-Cultural Gerontology*, 32: 303–21. doi: 10.1007/s10823-017-9332-6